Author's response to reviews

Title: Factors influencing dental care access in Jordanian adults

Authors:

Suhair R Obeidat (saobeidat@just.edu.jo)
Amani G Alsa’di (amani.sadi@yahoo.com)
Dafi S Al-Taani (dtaani@just.edu.jo)

Version: 4 Date: 6 October 2014

Author's response to reviews: see over
Dear Dr Morawska,

Re: Manuscript reference No. 1573952035135257

Please find attached a revised version of our manuscript “Factors influencing dental care access in Jordanian adults”, which we would like to resubmit for publication as a Research Article in *BMC Oral Health*.

Revisions in the text are shown using red colored font for additions. In accordance with the editor’s comments, we added a sentence of the funding sources in the acknowledgements. In the author contributions section, we clarified that all authors read and approved the final version of the manuscript, and we included the full unabbreviated version of ‘JUST’ in the ethics statement. We hope that the revisions in the manuscript and our accompanying responses will be sufficient to make our manuscript suitable for publication in *BMC Oral Health*. In the following page are our point-by-point responses to the editor’s comments.

We shall look forward to hearing from you at your earliest convenience.

Yours sincerely,

Suhair R. Obeidat, MSDH (USA), Instructor
Department of Applied Dental Sciences
Faculty of Applied Medical Sciences
Jordan University of Science and Technology
P.O. Box 3030
Irbid, Jordan 22110
Tel.: +962-2-7201000 Ext.: (26925)
Fax: +962-2-7201087
E-mail: saobeidat@just.edu.jo
Responses to the Editor’s comments:

1. Please provide funding sources in your acknowledgements.

Response:

- We provided the funding sources in the acknowledgements. The project was supported by the Deanship of Research at Jordan University of Science and Technology.

2. Please clarify that all authors read and approved the final version of this manuscript in your author contributions section.

Response:

- The final version of the manuscript was read and approved by all authors. We clarified that all authors read and approved the final version of the manuscript in the author contributions section.

3. In your ethics statement, please include the full unabbreviated version of ‘JUST’

Response:

- We included the full unabbreviated version of ‘JUST’ (Jordan University of Science and Technology) in the ethics statement.